08/17/2020 – EFOG Meeting
	Present: Nena, Bruce, Liz, Steve, Wendy, Ashley, Bruno

Nena
· Nothing major to report from Ops Meeting
· We have 5k to 8k left in our budget for this FY.  If you need something, let Nena know.  The item has to be delivered by 9/16/2020.
· 2 FC Interviews were held last week, there are more scheduled in the coming weeks
· In order to make the interviewee feel less uncomfortable, maybe only have the people asking the questions appear on video
· John Mazzio is helping HP during shutdown on various days
· Argonne Associates are not allowed on-site at this time and not sure when they will be allowed on-site
· Ashley has made a spreadsheet to coordinate the startup activities for when we begin running again on Sept 14th…CCCLs, QEW walk-throughs, etc.
· There is also the CCWP spreadsheet in Box
Wendy
· 25-ID did install the ratchet wall collimator
· Was able to confirm the lead brick stacking via photos
· The work should now be progressing on the FE
Ashley
· Sector 27 water line for leveling system supposed to be removed today but the work was already done and no pictures of the lead restacking had been sent to Wendy
· The lead is now being unstacked, pictures will be taken and sent to Wendy, and the lead will then be restacked
· Sector 18 FERDP work is done; labyrinths still open
· Need to confirm if the work at 20BM is still happening this week (this is work that was deferred from the previous shutdown)
· Sector 21 relocating PSS door control box; box is off; need to weld new brackets
· Sector 22 welding of brackets has not started yet
Steve
· Sector 30 work not yet started on removal of labyrinths on roof
· No new progress on the repair work at 31-ID
Bruno
· LTHW work now at Sectors 8 and 11; Sector 7 is essentially done; work seems to be taking a few days per sector
· The key for the 35ID ratchet wall was missing
· Someone from the cryo group had the key
· The logs have now been updated to reflect who had the key and when they had they the key
· Still need to order some O2 sensors (per Nena Bruno will spend less than 1k on this)
· Will communicate with John Mazzio regarding the calibration schedule of the monitors during shutdown
· Nena will look at the email from Ed Russell regarding the backup AES for 437
[bookmark: _GoBack]Liz
· GM/CA (Sector 23) TYSSR Tuesday, August 18th at 11am on BlueJeans
· Created a spreadsheet to track commissioning ESAFs for 2020-3.  It is located in the same folder as Ashley’s start up spreadsheet
· Per Nena’s chat, the ESAF table in Shift Log could use some attention
Bruce
· Still waiting for Yu to complete the User Office testing so that she can work on ESAF updates
